Author’s response to reviews

Title: The IGNITE (Investigation to Guide New Insight into Translational Effectiveness) Trial: Protocol for a Translational Study of an Evidenced-based Wellness Program in Fire Departments

Authors:

Diane L Elliot (elliotd@ohsu.edu)
Kerry S Kuehl (kuehlk@ohsu.edu)
Esther L Moe (moe@ohsu.edu)
Carol A DeFrancesco (defrance@ohsu.edu)
Linn Goldberg (goldberl@ohsu.edu)
David P MacKinnon (davidpm@asu.edu)
Jeanne Enders (endersj@pdx.edu)
Kim C Favorite (kimcfavorite@msn.com)

Version: 4 Date: 30 September 2010

Author’s response to reviews:

MS: 1007721211436220

Study protocol

The IGNITE (Investigation to Guide New Insight into Translational Effectiveness) Trial: Protocol for a Translational Study of an Evidenced-based Wellness Program in Fire Departments

Diane L Elliot, Kerry S Kuehl, Esther L Moe, Carol A DeFrancesco, Linn Goldberg, David P MacKinnon, Jeanne Enders and Kim C Favorite

Dear Editorial Team,

I have submitted a clean copy of the manuscript listed above to the website. I accepted all of the changes, except leaving changing the “&” in Oregon Health & Science University to “and,” and the reference to the website has been checked and updated in the manuscript.

We appreciate your ongoing assistance.

Sincerely yours,

Diane

Diane L. Elliot, MD, FACP, FACSM
Professor of Medicine
Division of Health Promotion & Sports Medicine
Oregon Health & Science University
3181 SW Sam Jackson Park Road (CR110)
Portland, Oregon 97239
(503) 494-6554
FAX (503) 494-1310
email: elliotd@ohsu.edu